

08-Dec-2020

Lian-Sheng Ma
Science Editor
Baishideng Publishing Group Inc

Manuscript NO.: 59354

Gastrointestinal-type chemotherapy prolongs survival in an atypical primary ovarian mucinous carcinoma: A case report

Name of Journal: World Journal of Clinical Cases

Dear Prof. Ma

Thank you very much for your letter and the review of our manuscript. We have carefully addressed the reviewers' comments and revised the manuscript as requested. Our point-by-point responses to the reviewers' comments are detailed below. We also highlighted in red the major changes made in the text.

We highly appreciate the reviewers for their valuable and concrete comments, which are very helpful for improving our manuscript. We hope that these changes will make the manuscript acceptable for publication in World Journal of Clinical Cases. If there are additional changes that we should make, please let us know. Thank you very much for your consideration.

Yours sincerely

Qi Wang
Department of Radiation Oncology
the Affiliated Hospital of Qingdao University, Qingdao266000, Shandong
E-mail: wqwhrmh@163.com

Response to Reviewer#1 :

Comment 1 : In the Abstract could be better defined the criteria employed in the choice of the type of chemotherapy.

Response: Thanks for your suggestion. We have added the criteria in Abstract section.

Comment 2: In the pathological history of the patients should be useful to report the indication for hepatic resection performed in the past and also some details of the pathological features of the lesion.

Response : We highly appreciate your constructive comments and suggestions. We have made correction according to the Reviewer's comments. The details of the pathological features were described in the part of History of past illness.